Author’s response to reviews

Title: Factors influencing the attention to home storage of medicines in China

Authors:
Yin Huang (share0122@126.com)
Lingjie Wang (1119845129@qq.com)
Changqing Zhong (zchqfly@126.com)
Shumin Huang (1948357482@qq.com)

Version: 4 Date: 06 Jun 2019

Author’s response to reviews:

Responses to reviewer/editorial comments on the manuscript entitled "Factors influencing the attention to the home storage of medicines in China" (PUBH-D-18-03866R3)

Dear editors and reviewers:

Thank you for your letter and for the editorial comments concerning our manuscript entitled "Factors influencing the attention to home storage of medicines in China" (PUBH-D-18-03866R3).

Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our research. We have studied comments carefully and have made correction which we hope will meet with your approval. The main corrections in the paper and the responds to the editorial comments are as follows:

Response to the editorial comments:

Editor Comments:
1. We note that the current submission contains some textual overlap with other previously published works, in particular:
and
This overlap mainly exists in the Background and Discussion sections. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase these sections to minimise overlap.

Response:
We have re-phrased these sections. Please see the Background and Discussion sections.

2. Please remove the funding information from the Acknowledgements and include it in the Funding section instead.
Response:
We have removed the funding information from the Acknowledgements and included it in the Funding section instead.

3. We note that you have included a footnote. Due to the nature of online publishing footnotes cannot be published. Please can you either move the information in the footnote to the main text or alternatively provide it in an ‘Endnotes’ section. You can include an ‘Endnotes’ section after the ‘Declarations’ section and before the References. Endnotes should be designated within the text using a superscript lowercase letter and all notes (along with their corresponding letter) should be included in the Endnotes section. Please format this section in a paragraph rather than a list.
Response:
We have included an ‘Endnotes’ section after the ‘Declarations’ section and before the References. Please see the ‘Endnotes’ section.

4. Thank you for uploading your cover letter, title page, manuscript with tracked changes, response to reviewer and STROBE checklist. As these files are no longer required at this stage of the editorial process, please remove them from your submission. Please also remove the STROBE checklist from the additional files list and re-name your additional files to reflect this. (e.g. Additional File 4 should become Additional file 3).
Response:
We have removed these files from our submission and re-named our additional files.

5. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication. Currently we note that not all of your additional files are cited within the text.
Response:
All the figures/tables and supplementary files have been cited within the text. Please see the revised manuscript and supplementary files.

6. Please upload your revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.
Response:
The revised manuscript is a single, final, clean version that without any tracked changes, comments, highlights, strikethroughs or text in different colours.